MA194 { Using Unix

Instructor: Dan Chaney (chaney@ms.uky.edu) Office hours: Mon 2-3, 606-257-2150

Class #1 - Tuesday, January 16, 1996 – Week 1

In-class hotline: (606) 323–3681 Community Colleges: Tues 9:00-9:50 & 10:00-10:50 PM EST UK/Lexington: Wed & Fri 9:00 PM EST, TCI Channel 16

URL: http://www.ms.uky.edu/~ma194

Mailing list: ma194-class@ms.uky.edu To join the mailing list, send a note to ma194@ms.uky.edu

Tonight’s Topics – Overview & Welcome to the Class – World Wide Web (Access) – Requesting an account (via WWW) – The MA194 Home page – The MA194 Mailing List – Homework! (Already?!?)

Homework

This week’s homework will be easy and fun. To get ready for the class, you need two things: 1 - Browser access to the World Wide Web 2 - Ability to send & receive electronic mail

So, this week’s homework is to:

1. Send email to ma194@ms.uky.edu.

If you want to join the mailing list, include that in your note!

2. Request an account via the MA194 Class Web page http://www.ms.uky.edu/~ma194

URLs of the week

The MA194 Home Page
http://www.ms.uky.edu/~ma194

Registration information

If you have questions concerning registration for this course, contact: Bill Verble 1-800-432-0963 ext 73377

Satellite info

KU Band Spacenet II Transponder 23 Channel 9 or 10

Broadcasts from 9:00 to 11:00 PM EST